Author's response to reviews

Title: Simultaneous Aurora-A/STK15 overexpression and centrosome amplification induce chromosomal instability in tumour cells with a MIN phenotype

Authors:

Laura Lentini (laura.lentini@unipa.it)
Angela Amato (amatoangela@unipa.it)
Tiziana Schillaci (tizianaschillaci@unipa.it)
Aldo Di Leonardo (adileon@unipa.it)

Version: 3 Date: 9 November 2007

Author's response to reviews: see over
Palermo, 9 November 2007

Dear Sirs,

thank you for your letter confirming that BMC Cancer will publish our manuscript:

“Simultaneous Aurora-A/STK15 overexpression and centrosome amplification are required to induce chromosomal instability in cells with a MIN phenotype” by Laura Lentini, Angela Amato, Tiziana Schillaci and Aldo Di Leonardo (MS: 7269406251453814)

We made the following changes to the format of our manuscript as requested:

Affiliation- affiliations were written in full and each author was liked to the corresponding affiliation with the use of superscript numbers.

Section headings- sentence case was used for all headings and sub-headings in the manuscript.

Abstract – Abstract was formatted as requested adding the Methods part. Abstract details were updated on the submission page.

Sincerely,

Dr. Aldo Di Leonardo
Dipartimento di Biologia Cellulare e dello Sviluppo
Università di Palermo, viale delle Scienze 90128, Palermo, Italy
Phone #: +39 091 6577340; Fax #: +39 091 6577347
E-mail: adileon@unipa.it